Author's response to reviews

Title: Unsuspected Pneumocystis pneumonia in HIV-seronegative patient with untreated lung cancer: a case report

Authors:

Cai Chuang (skinblack1966@yahoo.com.cn)
Xie Zhanhong (Drxiezhanhong@163.com)
Gu Yinyin (DrGuinyin@163.com)
Zhong Shuqing (Drzhongshuqing@163.com)
Zhong Nanshan (DrNanshanzhong@163.com)

Version: 2 Date: 27 June 2007

Author's response to reviews:

Dear editor:
Thank you very much for giving us another chance and providing your kind suggestions.
In the revised manuscript, first we reconstructed our manuscript following the guidance to authors. Secondly, we corrected a few mistakes in grammar and syntax. Thirdly, we cut off some redundancy to make the manuscript more concise.

Yours most sincerely


2007-6-27